Jet triggers in the new data

▷ Run 142646

- 2002 JAN 01, 00:28:39 - 14:36:36 - about 285k events - global CalMuon trigger version 3.10 ▷ LBN files

- stage1b - preliminary numbers that has to be tested - prescale factors does not correspond to the L1 trigger luminosity find by

lm access package

▷ L1 and L3 trigger bits mess - array L3bits in the EVENT branch in root-tuples is not indexed by L3 trigger bit but by L1 trigger bit number - each L1 jet trigger has at least two L3 triggers

Run 142646, no JES, JCCA

JES v1.0 and L1 trigger efficiencies

▷ the latest JES corrections (v1.0) ▷ L1 trigger efficiencies

- provided by Frank Filthaut (Jet/MET ID meeting in Dec 05, 2001)